



REVIEW HISTORY

Paper: **“The Center- Margin Dialectics in The Poetry of Emmanuel Fru Doh and Bill F. Ndi”**

Corresponding Author: Tem Edwin Nji

Email: njitem105@gmail.com

Doi: 10.19044/llc.v6no4a2

Peer review:

Reviewer 1: Durrell Nelson

Liberty University, USA

Reviewer 2: Blinded

Reviewer 3: Blinded

Published: 29.12.2019

